Author’s response to reviews

Title: A qualitative study of employees' opinions on establishing a generic call-centre

Authors:

Hilde Carin Storhaug (hilde-carin.storhaug@trondheim.kommune.no)
Sara Maed (sara.mead@ntnu.no)
Aslak Steinsbekk (aslak.steinsbekk@ntnu.no)

Version: 3 Date: 15 Sep 2017

Author’s response to reviews:

The title in the abstract is changed from Objectives to 'Background' both in the manuscript and in the editorial system.

In the Ethics approval and consent to participate section, the following text is included: 'Prior informed written consent was obtained from all the study participants.' and the title is changed to 'Ethics approval and consent to participate'.